 CHOHAN /03-Oct- 05

Week 40 starting 3 oct 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  3 Oct 05 : 10/3/2005 2:49:20 PM
	6
	A1
	MBBR3389
	Prep 5 Cool Down
	118.33

	11
	A2
	MBAL3400
	PT 14 Magnet MAPS
	295.44

	10
	B1
	SSS642
	PT 13.2 Resist. Meas.
	295.18

	7
	B2
	SSS167
	Prep 5 Cool Down
	100.57

	8
	C1
	MBAL3734
	Prep 5 Cool Down
	290.71

	9
	C2
	MBAL2064
	ICS 4 Final connection
	292.85

	3
	D1
	SSS091
	PT 6.8 Training 8
	13.24

	4
	D2
	SSS630
	PT 2 HV Cold Test
	1.91

	1
	E1
	MBBL2203
	PT 12 Warm Up
	6.64

	12
	E2
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	291.15

	5
	F1
	MBBL2204
	PT 5.1 1.5 KA Quench LF
	1.90

	2
	F2
	MBBR1257
	PT 12 Warm Up
	296.34


   Tempr. from TBSM18 site
 
   
Coordinator this week: V.Chohan; MTM support: 16-3602

16 complete Magnet Tests cycles 
 13 dipoles ( 1017 long-term stability tests with MM over 2 wks  )+ 3 SSS(without MM & SSW ) 

+no  ssss Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
Situation From Cryogenics:  J-P Lamboy or ACR Rep.
E2 CFB (leak) Repair whole Week this week?  ( 11 benches usage only)
 Special  Points :
· Special SSS on B1 : Disconnection & new ( prospects & plans – J Halik ) 
· Special SSS on D2 :Tests Plans - Marco B ?

· OP Requests to AB-CO

· AOB
 

	Magnets under Test from 25-09-2005 to 01-10-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAL1221 
	[image: image3.png]



	


	26-09-2005 
	43 h.
	29 h.
	15 h.
	13 h.
	9 h.
	01-10-2005 
	109 h.
	Stripping
	
	

	2
	TBA1
	MBAL2195 
	[image: image5.png]



	


	19-09-2005 
	24 h.
	60 h.
	53 h.
	18 h.
	10 h.
	26-09-2005 
	165 h.
	Stripping
	
	

	3
	TBA2
	MBBR1017 
	[image: image7.png]



	


	15-09-2005 
	30 h.
	21 h.
	248 h.
	16 h.
	12 h.
	29-09-2005 
	327 h.
	Stripping
	
	

	4
	TBB2
	SSS089 
	[image: image9.png]



	


	27-09-2005 
	30 h.
	18 h.
	14 h.
	10 h.
	23 h.
	01-10-2005 
	95 h.
	Stripping
	
	

	5
	TBB2
	SSS215 
	[image: image11.png]



	


	23-09-2005 
	36 h.
	14 h.
	24 h.
	10 h.
	12 h.
	27-09-2005 
	96 h.
	Stripping
	
	

	6
	TBC1
	MBAL3724 
	[image: image13.png]



	


	23-09-2005 
	22 h.
	24 h.
	23 h.
	14 h.
	12 h.
	27-09-2005 
	95 h.
	Stripping
	
	

	7
	TBC1
	MBBR3386 
	[image: image15.png]



	


	27-09-2005 
	14 h.
	38 h.
	12 h.
	14 h.
	23 h.
	01-10-2005 
	101 h.
	Stripping
	
	

	8
	TBC2
	MBAL2132 
	[image: image17.png]



	


	21-09-2005 
	36 h.
	22 h.
	18 h.
	17 h.
	9 h.
	25-09-2005 
	102 h.
	Stripping
	
	

	9
	TBC2
	MBBR3385 
	[image: image19.png]



	


	25-09-2005 
	29 h.
	21 h.
	25 h.
	17 h.
	8 h.
	30-09-2005 
	100 h.
	Stripping
	
	

	10
	TBD1
	SSS180 
	[image: image21.png]



	


	18-09-2005 
	38 h.
	50 h.
	145 h.
	11 h.
	35 h.
	30-09-2005 
	279 h.
	Stripping
	
	

	11
	TBE1
	MBAR3393 
	[image: image23.png]



	


	21-09-2005 
	36 h.
	22 h.
	13 h.
	14 h.
	15 h.
	25-09-2005 
	100 h.
	Stripping
	
	

	12
	TBE1
	MBAR3394 
	[image: image25.png]



	


	25-09-2005 
	26 h.
	23 h.
	35 h.
	25 h.
	17 h.
	30-09-2005 
	126 h.
	Stripping
	
	

	13
	TBE2
	MBBL2203 
	[image: image27.png]



	


	24-09-2005 
	28 h.
	34 h.
	23 h.
	15 h.
	14 h.
	29-09-2005 
	114 h.
	Stand By
	
	

	14
	TBF1
	MBAR1252 
	[image: image29.png]



	


	26-09-2005 
	28 h.
	26 h.
	14 h.
	15 h.
	15 h.
	30-09-2005 
	98 h.
	Stripping
	
	

	15
	TBF1
	MBBL1198 
	[image: image31.png]



	


	22-09-2005 
	27 h.
	21 h.
	22 h.
	17 h.
	15 h.
	26-09-2005 
	102 h.
	Stripping
	
	

	16
	TBF2
	MBBL2212 
	[image: image33.png]



	


	24-09-2005 
	31 h.
	35 h.
	39 h.
	16 h.
	7 h.
	29-09-2005 
	128 h.
	Stripping
	
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL1221
	11573 A.
(D2-L)
	12209.5 A.
(D1-L)
	12601 A.
(D2-L)
	[image: image35.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL2195
	11100 A.
(D1_U)
	11723 A.
(D2-L)
	12276 A.
(D2-L)
	12663 A.
(D1-L)
	[image: image36.png]



	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBR1017
	12507 A.
(D@-L-S13)
	12850 A.
[image: image37.jpg]



	12850 A.
[image: image38.jpg]



	12850 A.
[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS089
	12498 A.
(QE-34)
	12853 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	5
	SSS215
	12054 A.
(QI-34)
	12543 A.
(QI-34)
	12796 A.
(QE-34)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	6
	MBAL3724
	10102.4 A.
(D1_L)
	11920 A.
(D1 - L)
	12215 A.
(D1 - L)
	12473 A.
(D1L)
	12714 A.
(D1L)
	12849 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-
	
	

	7
	MBBR3386
	12588.6 A.
(D2-L)
	12800 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	8
	MBAL2132
	11047.3 A.
(D1_U)
	11901.5 A.
(D1_U)
	12515 A.
(D1_L)
	12850 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	9
	MBBR3385
	11491 A.
(D1-U)
	12843 A.
(D2-U)
	12850 A.
[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	10
	SSS180
	12210 A.
(QE_12)
	12370 A.
(QE_12)
	12644 A.
(QE_34)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	11
	MBAR3393
	10499 A.
(D2_U)
	12045 A.
(D1_L)
	12852 A.
(D1-U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	12
	MBAR3394
	8709 A.
(D2 - U)
	10267 A.
(D1-U)
	12170 A.
(D2 - L)
	12245 A.
(D2-L)
	12684 A.
(D2-U)
	12850 A.
[image: image48.jpg]



	-
	-
	-
	-
	-
	-
	-
	
	

	13
	MBBL2203
	9872 A.
(D1-L)
	11432 A.
(D2-L)
	12681 A.
(D1-U)
	12732 A.
(D2_L)
	[image: image49.png]



	-
	-
	-
	-
	-
	-
	-
	-
	
	

	14
	MBAR1252
	11213 A.
(D2_U)
	12854 A.
(D2_L)
	12850 A.
[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	15
	MBBL1198
	11809 A.
(D1 - U)
	12212 A.
(D1-L)
	12380 A.
(D2-U)
	[image: image51.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	16
	MBBL2212
	11884 A.
(D1_U)
	12559.8 A.
(D1_L_S01)
	12850 A.
[image: image52.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
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Shows no SSSS completed last week on D2 or B1  ;HOWEVER,  S180 had long cold tests period  due to COHABITATION  with Special SSS on D2   &   somewhat LONG COOLDOWN due to “forgotten vacuum barrier”
Memorandum

To:
AB-CO Responsible(s) for SM18 Applications & Support

CC:
L.Walckiers AT-MTM

From:
SM18 Operation Team

Date:
10/4/2005
Re:
Version Changes in online Applications Software, Applications for SSS Testing etc  

1) SUNMTA33 (B Cluster) and SUNMTAxx (D Cluster) have a new version of AQA (vs 6.31)  which replaces the previous one ( v.6.21), but this version is installed only on the  local floor-level SunWS, so it is not possible with this version to check and validate the data. This new version is an improvement and permits to see the 4 quench heaters curves for a quadrupole ( IR- SSS’s and  Arc-SSS’s ) because previously these were  not visible & were hidden by the application.  Upstairs in the Control Room, we have the previous version which permits full checks & validation, but this version is not correctly equipped to test all Arc or IR SSS’s. We have to remind all concerned that it is the SM18 Op team which checks the goodness of these signals by eye to ensure the quench heaters integrity,  at low current as well as during higher (training) currents. In other words, it is not an automatic process as the name AQA implies. Hence it is important that these signal observations and validation tools are correctly provided and installed appropriately. 

 2) The application Instrumentation Analysis Procedure (IAP) for warm temperature tests (at the beginning & at the end of a magnet tests cycle) is not working for the voltage taps reading for Arc-SSS’s. This has been requested since early this year, (April 05); For OP team, the IAP application for SSS ‘s is seriously lacking the homogeneity with Dipole Testing and causes confusion amongst newcomers in Operation because of lack of similarities and usage. The OP team has been criticized several times this year for SSS Testing & IAP (including reasons for ICS re-open & re-connections); secondly, the rhythm of Arc-SSS testing has gone up dramatically in 2005 as shown by tested magnets statistics. It is extremely frustrating and de-motivating for the 24 hr operation staff that IAP version still remains the same after all these months  and does not follow the AQA improvements & evolution as mentioned  in point (1) above.

	Date
	Bench
	Problem
	Type
	Application/
soft
	Action/advise

	Tuesday_27_A
	TBE2

	Unable to launch training quench on MB2203.

Potaim card was permanently triggered.


	MTM
	
	Federico changed a potaim card.

	Tuesday_27_N
	TBE2

	HWREC error: the database did not recognized the card.


	MTM
	
	Federico solved the problem by declaring a potaim-card in MTF

	Tuesday_27_N
	TBE1

	MB3394: During the slow power abort check at 1000A an error occurred on one  potaim card .


	MTM
	
	Federico advised to proceed with further tests.

	Wednesday_28_M
	TBF2

	LHC cycle on MBAL2212: Rotation speed above the limits.
	MTM
	
	TC advised to proceed with the measurements: origin of the problem may be come from the TRU.

	Wednesday_28_M
	TBE2

	Suspected leak on MB2203. Magnet or CFB?


	CRI/ACR
	
	Magnet put on stand_by till results of investigation by ACR..

	Wednesday_28_M
	TBA2

	MB1017: T reading is erratic.:


	MTM
	
	NC opened. Magnet signed in stripping as for TC instruction.

	Wednesday_28_A
	TBC2

	The rorary switch  is not switching to CC position from C2.


	AB/PO
	
	Federico called. He contacted AB/PO person on stand_by.

	Wednesday_28_A
	TBB1

	SSS642 training quench at 4.55 K: DirVT voltage peak Exceeded 150V 
	MTM
	AQA
	

	Wednesday_28_A
	TBF2

	Small spike in one quench heater signal on MBAL2212.
	MTM
	AQA
	Checked by Federico. Advised to continue.

	Thursday_29_A
	TBD1

	Resistance &voltage measurements at warm after cold test: The sextupole voltage taps have incorrect measurement values.
	MTM
	
	TC informed. NC opened.
On TC's advise the magnet has been signed for stripping as the sextupoles have been trained to the ultimate value of 600A

	Thursday_29_N
	TBA2

	HV test not possible with rack MY93 because a connector is damaged.
	MTM
	
	The HV test was performed with the rack MY92. Action Richard?

	Friday_30_M
	TBB2

	SSS089: validation of the data with AQA 6.31  not possible for the two filesG0509292225.ha01 and G0509300156.a01
	AB/CO
	AQA
	Federico informed. He advised to :-go on with the tests
-validate the data with the AQA in another bench.

Problem was transmitted to B.Khomenkho.

Reason: not a problem of version but of authorisation. Data validation was permitted only with sun MTA24

	Friday_30_A
	TBA1

	MB1221: T_reading not correct during SPA at 1000 A. Then the reading was OK.
	MTM
	
	NC opened. Federico informed.

	Friday_30_A
	TBE2

	Magnet MBBL2203 removed from TBE2 with suspected leak will be tested for leak on bench TBE1. 
	ACR
	
	TC informed that the magnet will be connected, cooled down and checked for leaks. No tests are to be carried our on this magnet except a quench required for warming up. The magnet has been put in TBE1.

	Friday_30_A
	TBD1

	IAP at warm on SSS091 could not be launched: Config. files in HWREC were missing.
	MTM
	HWREC
	Federico completed the config,files. Test re-launched and then OK.

	Saturday_01_M
	TBF2

	Communication between PLCart_and cryo_supervison not working.
	MTM
	
	Solved by switching on and off the anti-cryostat heating control power supply.

	Saturday_01_M
	TBE1

	Lyre test on the MB 2203:  The voltage did not drop when the power supply was switched on (12V, 2A). OK above 12V. Magnet resistance measured at 6.06ohm.
	MTM
	
	OK after discussing with Federico..

	Saturday_01_A
	TBF2

	E-traveller displayed an error message :Microsoft JET Database Engine error '80040e14'
Syntax error (missing operator) in query expression 'CONDITION ='. /tbsm18/_private/production.asp, line 518
	
	
	

	Saturday_01_A
	TBC1

	MB3394 put by mistake in the bench. It was declared stripping coming from the E1 bench.
	ICS
	
	Magnet removed by ICS and changed with the MB3734.

	Saturday_01_N
	TBB2

	IAP at warm after connection on SSS167. LF could not be armed because of an interlock  on the V_clamp_D
	ICS?
	
	OP team tried to contact ICS to check the connection.

	Sunday_02_M
	TBB1

	SSS089: HV test after cold test .MQ vs ground had a leak current of 1mA.
	MTM
	
	Federico asked to check with a voltmeter the leak current.
The connector P22 was not disconnected during test. After disconnection test OK.

	Sunday_02_A
	TBC1

	Hardware configuration not correct: no TT821 signal during IAP@warm
	CRI
	
	Federico called. He found that the temperature sensor is not in the database for the MB3734. HWREC off and test program pursued.

	Saturday_01_N
	TBB2

	IAP at warm after connection on SSS167. LF could not be armed because of an interlock  on the V_clamp_D
	ICS?
	
	OP team tried to contact ICS to check the connection.











